Author's response to reviews

**Title:** Incidence of chemotherapy-induced amenorrhea associated with epirubicin, docetaxel and navelbine in younger breast cancer patients

**Authors:**

Wen-Bin Zhou (zhouwenbin_1984@qq.com)
Hong Yin (77077977@qq.com)
Xiao-An Liu (liuxiaoan@126.com)
Xiao-Ming Zha (zhaxm@medmail.com.cn)
Lin Chen (hxchenlin@yahoo.com.cn)
Jun-Cheng Dai (stata9.2@163.com)
Ai-di Tao (enbens@njmu.edu.cn)
Ling Chen (rainbowlyn@gmail.com)
Jing-Jing Ma (honeymjj@hotmail.com)
Li-Jun Ling (pangpangzheng@sina.com)
Shui Wang (ws0801@hotmail.com)

**Version:** 5  **Date:** 10 June 2010

**Author's response to reviews:** see over
Dear Editorial Production Team:

Thank you very much for your letter dated on June 9, 2010. A few changes were made.

1) We changed “Wen-Bin Zhou and Hong Yin contributed equally to this work” to “Wen-Bin Zhou, Hong Yin and Xiao-An Liu contributed equally to this work” in Page 1.
2) We changed the color from red to black in Line 2-4, Page 4.
3) We added the illustration (*Reference category) to Table 3.

Thank you very much.

Best regards

Zhou Wenbin, MD.
Department of Breast Surgery
the First Affiliated Hospital of Nanjing Medical University
Email: zhouwenbin_1984@foxmail.com